Author's response to reviews

Title: Effectiveness of joint mobilisation after cast immobilisation for ankle fracture: a protocol for a randomised controlled trial [ACTRN012605000143628]

Authors:

Chung-Wei C Lin (c.lin@usyd.edu.au)
Anne M Moseley (amoseley@mail.usyd.edu.au)
Kathryn M Refshauge (k.refshauge@fhs.usyd.edu.au)
Marion Haas (marion.haas@chere.uts.edu.au)
Robert D Herbert (r.herbert@fhs.usyd.edu.au)

Version: 2 Date: 15 May 2006

Author's response to reviews:

The Editor

BMC Musculoskeletal Disorders

Dear Sir/Madam,

Thank you for accepting our manuscript for publication in BMC Musculoskeletal Disorders. We have made formatting changes as requested.

Underlining - all underlings have been removed in the text

Figure title - the titles have been removed from the figures. Figure legends in the text after the reference section.

Figures - we have looked at our figures and have decided not to crop the figures.